**Author's response to reviews**

**Title:** Patterns and socioeconomic influences of tobacco exposure in tobacco cultivating rural areas of Yunnan Province, China

**Authors:**

Le Cai (caile002@hotmail.com)
Xinan Wu (wxnan@hotmail.com)
Abhinav Goyal (agoyal4@emory.edu)
Yuntao Han (hytao@yahoo.com)
Wenlong Cui (Cwlong@yahoo.com)
Xia Xiao (xxia@yahoo.com)
Jianhui He (hjhui@hotmail.com)
Keying Zhao (zkying@hotmail.com)
Ying Song (sying@hotmail.com)
Feng Jiao (jfeng@yahoo.com)

**Version:** 2  **Date:** 23 March 2012

**Author's response to reviews:** see over
Cover Letter

To editors of BMC Public Health

Dear Sir or Madam:

I am submitting the revised manuscript “Patterns and socioeconomic influences of tobacco exposure in tobacco cultivating rural areas of Yunnan Province, China” (MS: 9906752556931176) to BMC Public Health. I have addressed my responses to the comments in the revised manuscript, and given a point-by-point explanation to the concerns in this cover letter.

The revisions/additions of the content are underline.

Best wishes.

Cai Le

School of Public Health

Kunming Medical University, Kunming, Yunnan province, China

e-mail: caile002@hotmail.com
Some changes have been made:
1. Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Done as what you suggested. Please see Page 6, second Paragraph.

2. Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

Done as what you suggested. Please see Page 6, first Paragraph.

3. Please provide us a copy of your questionnaire administered as an additional file.

Ok, no problem.